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Data Taking
   



   

   

   

   

   

    

   

   

 






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Significant Events and Accesses
● Tuesday, the L2 Trigger control computer was replaced.

– We expect this to solve our downtimes related to this issue.
● Thursday, Toroid trips during proton load.  Recovered before store.

● Friday, we made an opportunistic access.

– We recovered a few muon channels and preemptively replaced a CAL BLS power 
supply.

● Friday, The D0 clock was adjusted for seasonal changes.

● Friday evening a problem developed with 10% of the axial portion of the Fiber Tracker.

– A 30 minute access was requested.
● Saturday morning, the access was granted.

– The sequencer board we thought was the problem turned out to be OK.  It took an 
additional 40 minutes to locate and replace a faulty AFE board.

● The problem is still not completely understood and is still around at a reduced rate.
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On the morning of Wednesday,
7 October, D0 reached 7 fb­1

in delivered luminosity. 

We wish to thank the AD for 
the Luminosity and congratulate 
them on reaching this milestone.


